Author's Response To Reviewer Comments

Clo<u>s</u>e

We have thoroughly addressed all reviewers' comments pertaining to our manuscript "A probabilistic multi-omics data matching method for detecting sample errors in integrative analysis". Detailed responses are listed in the "Response to reviewers comments" letter.

Clo<u>s</u>e